Author's response to reviews

Title: Parents education level and caries status in children with an immigration background in Austria

Authors:

Barbara Cvikl (barbara.cvikl@meduniwien.ac.at)
Gertraud Haubenberger-Praml (gertraud@haubenberger.at)
Petra Drabo (petra.drabo@meduniwien.ac.at)
Reinhard Gruber (reinhard.gruber@zmk.unibe.ch)
Andreas Moritz (andreas.moritz@meduniwien.ac.at)
Andrea Nell (andrea.nell@meduniwien.ac.at)

Version: 2 Date: 10 September 2013

Author's response to reviews:

Dear Editors,

Thank you for the notification. Regarding the requested ethics statement we added the following statement to the Methods section:

The study was approved by the ethics committee of the Medical University of Vienna. The request for an ethics commission number was not considered necessary by the ethics committee. The epidemiological investigation was carried out after informed consent.

Sincerely,

Reinhard Gruber and Barbara Cvikl